Author’s response to reviews

Title: Association between tumour infiltrating lymphocytes, histotype and clinical outcome in epithelial ovarian cancer

Authors:

Fiona James (jamesfr@hotmail.co.uk)
Mercedes Jiminez-Linan (mercedes.jimenez-linan@addenbrookes.nhs.uk)
Jennifer Alsop (ja503@medschl.cam.ac.uk)
Marie Mack (mbm43@medschl.cam.ac.uk)
Honglin Song (hs310@medschl.cam.ac.uk)
James Brenton (James.Brenton@cruk.cam.ac.uk)
Paul Pharoah (pp10001@medschl.cam.ac.uk)
Raza Ali (Raza.Ali@cruk.cam.ac.uk)

Version: 2 Date: 02 Aug 2017

Author’s response to reviews:

Editor Comments:

1. Please remove the cover letter from the file inventory, as it is no longer needed at this stage of the editorial process.

Done

2. Please list the full names, institutional addresses and email addresses for all authors on the title page.

Done

3. Please rename the "Introduction" section to "Background" and "Materials and Methods" to "Methods" in order for the manuscript format to meet the journal guidelines. Please include a section in the manuscript titled Conclusions immediately after the Discussion section. This
should state clearly the main conclusions and provide an explanation of the importance and relevance of the study reported.

Done

4. Please include a working link to the data repository listed in the Availability of Data and Materials statement: The current link is "DOI Not Found."

Link now works

5. We recommend that you adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. Please see http://www.biomedcentral.com/getpublished/editorial-policies#authorship for more details.

Done

6. In the Funding section, please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Done - statement "The funders of this project had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript" added to that section